CENWP-OP-B                                                                                                                                          22 July 2003

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 29 of 2003, which covers the period 13 through 19 July 2003.

1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 132.9 to 194.1 kcfs.  Daily project tailwater elevation averaged 10.8’ to 15.0’ msl.  Daily project forebay elevation averaged 74.6’ to 76.3’ msl.  River temperature averaged 65 to 67° F.  Daily turbidity measured 5.0’ to 6.7’ on the secchi disk.

1.2 Spill Bays 1 and 18, which supply attraction flow for Cascade Island and Bradford Island fishways respectively, are included in the spill pattern.  Spill averaged 86.9 to 97.2 kcfs.  

1.3 Unit Operation

1.3.1 Unit 1 remains out of service for turbine rehab.

1.3.2 Unit 7 was out of service from 0430 on 14 July until 1518 on 17 July for headgate removal.  

1.3.3 Unit 8 was out of service from 0430 on 14 July until 1518 on 17 July for headgate removal.

1.3.4 Unit 9 returned to service at 1518 on 17 July.   Unit 9 has no STSs, it will be run last on/first off.

1.3.5 Unit 10 was out of service from 0430 on 14 July until 1518 on 17 July for headgate removal.

1.3.6 Unit 15 was out of service from 0330 until 1724 on 14 July for annual maintenance.

1.3.7 Unit 16 was out of service from 0336 until 1642 on 15 July for annual maintenance.

1.3.8 Unit 18 was out of service from 0330 until 1824 on 16 July for annual maintenance.

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.   

1.5 Gatewell drawdowns are measured on Wednesdays.

2.  ACTIVITIES. 
2.1 AWS Closures.  No AWS valves were closed this reporting week.

2.2 STS/VBS Inspections.  No STS/VBS inspections were conducted this reporting week.

2.3 Dewatering and Fish Salvage.  No dewatering or fish salvage activities this reporting week.

3.  RESEARCH.

3.1 CRITFC.  Sampling of chinook, sockeye, and coho continues until 31 October.  Reported one dead chinook.  See misc 5.9
3.2 NMFS Lamprey.  Trapping, tagging and tracking continue.

3.3 OCFWRU.  Fish collection concluded this week. 

3.4 ODFW Steelhead.  Trapping continues.  Reported one dead chinook.  See misc. 5.10
3.5 PSMFC at DSM1.  Continuous PIT tag interrogation continues.   See misc. 5.6
3.6 PSMFC at the SMF.  One kelt was observed stuck on the separator bars; one steelhead and two smolts slid over the bars.  GBT exams continue.

3.7 University of Idaho.  Adult tracking tagging and tracking continue.  

3.8 U of I Transportation Study.  Evaluation of Snake River fall chinook will continue until 15 August.

3.9 University of Washington.  Northern pikeminnow tagging and tracking from May through July.  

3.10 WDFW Fish Counting.  16-hr visual counts at Washington Shore and Bradford Island.

3.11 WES/Battelle/BAE systems Juvenile hydroacoustic monitoring.  Gap loss monitoring continues until 21 July.

4.  FISHWAYS 
4.1 Project biologists inspected 13 through 19 July this reporting week.

4.2 Fish Passage Plan Observations.

	DATE
	LOCATion
	FPP Deviation
	CAUSE
	RESPONSE
	TR/WAR

	All week
	PH1 PCC
	Velocity <1.5fps
	Meter in slowest part of channel
	No response at this time.
	

	All week
	PH2 PCC
	No velocity measure.
	Meter is missing
	See misc. 5.1
	

	All week
	PH1 ITS
	Chain gates 7A and 10C are closed.
	Headgate delivery.
	4B & 6B opened.  See misc. 5.4
	

	7/17
	Weir 38
	Reading >or <1.3’ +/- .1’
	Unknown
	
	

	7/14-15
	WA Shore PCC
	SUE, SDE differential <1.0’
	Faulty differential readout
	SA-24 board calibrated
	TR#48388


4.3 Fish Ladder Temperatures.  Measurements are taken three weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.  Temperatures are in Kelvin.  http://students.washington.edu/kyle/temptable.jpg   (295.37K = 22.22C= 72F)

	Date
	Wash. Shore
	Bradford Is.
	AFF exit

	7/13
	293.3
	239.2
	293.3

	7/14
	293.4
	293.4
	293.4

	7/15
	293.5
	293.5
	293.5

	7/16
	293.5
	293.5
	293.5

	7/17
	293.7
	293.6
	293.6

	7/18
	293.9
	293.8
	293.9

	7/19
	294.0
	294.0
	294.1


4.4 Sensor Calibration.  Calibration was checked on 15 July this reporting week.

4.5  Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week. 

5. MISCELLANEOUS

5.1 Divers discovered the PH2 velocity meter was missing from the collection channel.  A new meter will be installed during the mid season dives scheduled for 21 and 22 July.

5.2 As of 28 June, the Project will have a daytime spill of 75 kcfs.

5.3 Orifice 18A-N remains in local and open due to a bad proximity switch.

5.4 Due to headgate delivery, chain gates 7A and 10C were closed on 10 July.  To compensate, 4B and 6B are opened to 72’msl and 72’ msl respectively.

5.5 In a 10 July teletype the Project was given a hard constraint maximum tailwater of 16’ until 01 September.  This is for the B2CC work at PH2.  

5.6 As DSM1 returned to auto around 0948 on 14 July.

5.7 In a 14 July teletype, approval to exceed a maximum four foot increase in tailwater elevation in a 24 hour period was permitted until 2400 on 14 July.

5.8 In a 15 July teletype, the maximum tailwater constraint overrules spill to the 120% gas cap if spill raises tailwater over 16’.

5.9 At 1200 on 15 July CRITFC reported one dead clipped chinook at the AFF.  The fish appeared to have healthy gills and no signs of injury or fungus.  There was what looked to be a paper punch hole in the caudal fin.  CRITFC believes the fish expired from stress due to warming water temperatures.

5.10 At 0830 on 16 July, ODFW reported one dead unclipped chinook at the AFF.  The fish jumped from the researcher’s grip before being released into the brail pool for recovery.

5.11 FFU and the Rangers planted native shrubs and trees around Bass Lake on 16 July from 0930 until 1200.  This is part of a larger plan to do restoration work at Bass Lake.

5.12 On 17 July STSs were installed in Unit 3 to clear the deck for crane rail modifications set to start 01 August.  

5.13 TIES were removed from units 14-16 on 17 July.

5.14 Due to the entrance level sensor reading low at B-Branch, FV4-4 was in manual from 1510 on 19 July until 0100 on 20 July to keep the fishway in criteria.

JAMES R. MAHAR, P.E.

Operations Manager
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